Author’s response to reviews

Title: An immediate effect of PNF specific mobilization on the angle of trunk rotation and the Trunk-Pelvis-Hip Angle range of motion in adolescent girls with double idiopathic scoliosis – a pilot study

Authors:
Agnieszka Stępień (orthosas@wp.pl)
Krzysztof Fabian (thruman@o2.pl)
Krzysztof Graff (graffk@wp.pl)
Małgorzata Podgurniak (mpodg@wp.pl)
Andrzej Wit (andrzej.wit@awf.edu.pl)

Version: 3 Date: 08 Apr 2017

Author’s response to reviews:

Dear Reviewers,

Once again we would like to thank you for your support and time. Your opinions gave us a possibility to learn and improve our manuscript.

Reviewer reports:

Reviewer #1: Table 2: it's not necessary to have both SD and SE, chose one of the two.

--> The table was corrected
Table 4: still difficult for readers to understand. Again, I suggest the authors to calculate median or mean according to the way data are distributed and summarize your results. It seems the data are already presented in table 2....if this is the case I suggest to remove table 4.

--> Table 4 and accompanying text were removed from the manuscript.

Discussion

It's better but probably still too long with respect to the limited findings of the paper.

Line 246-247: was the change before or after the treatment? Unclear....

--> Line 246-247 was corrected. Thank you, it was a mistake. A few sentences was removed from the discussion to short it.

Reviewer #4: The paper is much better now. It can be a candidate for publication.

--> Thank you very much.